Bilevel Programming for Manufacturers Operating in an Omnichannel Retailing Environment Vishal Kumar Gupta, Shrinath Dakare, Kiran Jude Fernandes, Lakshman S. Thakur, Manoj Kumar Tiwari Abstract: This paper presents a joint production, pricing and inventory control problem that can help manufacturers manage revenue in an omnichannel environment. The paper puts forward a multi-period Stackelberg game between a manufacturer and a retailer, assuming the former to be the leader. The leaderfollower game is formulated as a mixed-integer non-linear bilevel optimization problem wherein both players seek to maximize their respective profits. The manuscript envisions an omnichannel retailing environment where online, offline, direct and drop-shipping channels coexist. It then investigates how enabling showrooming and webrooming on the drop-shipping channel account can affect supply chain profitabilities. Analyses suggest that when customers are aided with showrooming and webrooming services and are allowed to place drop-shipping orders from retailers' stores, additional profit can be generated in the supply chain. To solve the proposed bilevel optimization problem, we investigated a) single-level reduction using Karush-Kuhn-Tucker (KKT) conditions and b) hierarchical optimization technique based on Simulated Annealing and Randomized Decomposition Solver. To assess the efficacy of the solution techniques, a comparative analysis was carried out. Thereafter, with the aid of numerical experiments and sensitivity analyses, the paper draws key managerial insights for manufacturers. Keywords: Retailing, price optimization, omnichannel, bilevel programming, Stackelberg game Managerial Relevance Statement: The manuscript is relevant to the manufacturers who operate direct

channels in omnichannel settings. The paper highlights that manufacturers often lack pricing and revenue management solutions in their operations; such solutions would prove beneficial in mitigating the impact of demand and production mismatch. Therefore, the paper proposed joint production, pricing and inventory control solution for the manufacturers of seasonal products.

The paper asserts drop-shipping service as a means of cooperation between manufacturers and retailers who otherwise are non-cooperative decision-makers in decentralized supply chains. It numerically demonstrates that drop-shipping can not only serve as an omnichannel service but can also increase supply chain's profitabilities. However, a drop-shipping channel can also cannibalize the direct channel. Numerical experiments suggest that having a drop-shipping channel can bring positive changes to supply chain profitabilities in the case when production capacity is not sufficient. Whereas, in the case when production capacity is more, drop-shipping channel may not bring any significant change in the supply chain profitabilities. Finally, the results suggest that an increase in popularity of a drop-shipping channel can benefit the retailer more than the manufacturer.

I. INTRODUCTION

It is now common for retailers (particularly in fashion apparel and consumer electronics) to use pricing and revenue management. Revenue management helps influence demand and mitigate the effect of limited supply, limited inventory and/or finite selling seasons [1]. To do so, the point-of-sales data is intelligently processed to understand consumers' price sensitivity, preferences, and market trends. These insights are then used to actively influence consumer purchase decisions by changing the posted price of the product. Such active demand management (in contrast to passive management of inventory) reduces demand and supply mismatch, thereby yielding a better return on investment [2].

In 2020, when the entire world was dealing with the Covid-19 pandemic, Target Corporation (an American retailer) was able to grow its sales by 145% thanks to its newly adopted omnichannel fulfillment strategy [3]. In India, brands like Pepperfry, Zivame, Van Heusen, and Adidas have already introduced their omnichannel strategies. For example, Adidas' endless aisle and virtual footwear wall allow customers to search and order items that are not in stock at the physical stores [4]. A 2020 report by McKinsey states that companies that can provide omnichannel personalization can achieve 5–15% revenue growth [5]. The report also suggests that because more than 80% of total sales occur at physical locations, providing better offline personalization is therefore vital for growth. In the following paragraphs, we discuss how omnichannel retailing became so ubiquitous.

Thanks to the internet, information technology, and competitive third-party logistics services, in 2012 online retailing became the fastest-growing retail sector in the United States [6]. Facing competition from e-commerce giants like Amazon, traditional retailers like Walmart, Costco Wholesale, Tesco and Metro added online channels to their channel mix. Likewise, e-retailers like Fab.com, Gilt.com and JD.com also transformed themselves into multi-channel retailers by incorporating brick-and-mortar stores [7]. At the same time, manufacturers like Dell, Nike and Apple started operating their own direct online channels. Although direct channels were initially for informational and sales support purposes only [8], they have

become a means to balance the increasing power of multi-channel retailers and to improve supply chain performance [7].

Multi-channel retailing emerged as channels were progressively added to retailing portfolios. However, the abundance of retail channels and the omnipresence of the internet via smartphone also allowed consumers to search multiple channels concurrently. As per a report by McKinsey, in 2012 alone, about 20% of customers used mobile phones for product research—of whom about 40% did so while they were in the store [8]. Customers intentionally moved from one channel to the other to maximize their overall utility [9]. Specifically, when a customer moves from a physical store to an e-commerce website, that behavior is known as "showrooming" [6]; in contrast, when a customer moves from an online channel to a physical store, it is known as "webrooming" [7]. These customer activities collapsed the boundaries between multiple retail channels. A price change in one of the retail channels now affects the demand generated at all other consumer touchpoints, a phenomenon commonly known as "cross-channel demand substitution" [10]. Moreover, realizing the importance of providing a superior customer experience, many multi-channel retailers started deliberately integrating their retail channels—what we know as "omnichannel retailing." Although different firms can have different omnichannel strategies, the underlying premise of omnichannel retailing remains the same, which is to integrate multiple available channels and thereby provide a superior customer experience.

A. Motivation

Observing the need for revenue management in omnichannel retailing, researchers recommended that retailers upgrade their pricing solutions to incorporate multi-channel interdependencies and crosschannel demand substitution [11], [12]. The need for price optimization and revenue management arises whenever a fixed and perishable set of resources are sold to a population that is sensitive to prices, and since outlets of both retailer and manufacturer face this problem, they are both free to incorporate pricing and revenue management solutions into their businesses.

Manufacturers who are actively adopting direct channels lack pricing and revenue management solutions in their operations; such solutions would prove beneficial in mitigating the impact of demand and production mismatch. It is not uncommon for a product to have short selling seasons [13] in which

manufacturers are constrained by their limited production capacities [14]. Such industries include clothing, pharmaceuticals, toys, gifts, and firecrackers. During a selling season, there is a certain starting inventory as well as a limited production capacity, as production cannot be easily ramped up or down. This challenge calls for active demand management by these manufacturers to achieve a better return on investment. Furthermore, a revenue management solution must consider cross-channel demand substitution if the manufacturer is operating in an omnichannel retailing environment. The objective of this paper is to develop a joint production planning, pricing and inventory control solution for a manufacturer operating a direct channel in such an environment. Furthermore, the paper draws researchers' attention toward a relatively unexplored area where pricing and revenue management solution must be developed for manufacturers considering demand substitution and pricing competition from retailers' online/offline channels.

B. Stackelberg Game

While developing a pricing and revenue management solution for manufacturers, we must assume that the retailer will also be using price optimization to maximize their profit. During a selling season, a retailer seeks to maximize its profit by effectively pricing the product while considering the demand, wholesale price and fulfillment costs. Essentially, manufacturers must also price the product depending on anticipated demand, available inventory and limited production capacities [15]. Due to cross-channel demand substitution, a price change by the retailer can also affect the demand in the manufacturer's channels, and vice-versa [10]. Usually, revenue management models ignore channel substitution and assume that price affects demand only in their respective channels. However, cross-channel demand substitution cannot be ignored in omnichannel retailing. Since customers can easily evaluate prices and availability of the product in multiple available channels, neither retailer not manufacturer can set their price independently. This leads to a multi-period leader–follower Stackelberg game between manufacturer and retailer, wherein the manufacturer is traditionally seen as a leader.

Here, we present some of the research where similar scenarios are documented. In [14], the authors presented a bilevel problem where the retailer determines selling prices and advertising expenditures while the manufacturer optimizes wholesale price and cycle time for the products. Since the demand was non-

linearly influenced by pricing and advertising expenditures, the formulated problem was solved metaheuristically using the imperialist competitive algorithm. The authors proposed a modified assimilation strategy along with a diversification approach to improve the performance of the given algorithm. The focus of the paper was on the methodology rather than on managerial insights. Authors in [16] investigated a decentralized production-distribution supply chain problem and presented a nonlinear bi-level programming problem that reflected Stackelberg games between manufacturers and distributors. In that paper, the authors used a hierarchical solution algorithm that combined genetic algorithm and particle swarm optimization to solve the formulation. The authors concluded that a leader always earns more than the follower and highlighted the finding that cooperation between the players can increase the supply chain profit. However, the paper did not explain how cooperation can be formulated and solved. In [17], the authors presented a competitive supply-chain network design problem. Authors assumed that rivals enter a new market and design their network to establish distribution centers. Both players sought to maximize their market share and minimize their costs using bilevel programming. The formulated problem was a nonconvex mixed-integer nonlinear problem that was convexified and solved using a two-step heuristic to attain global optimal. Authors found that when a rival "supply chain" is expected to enter a new market, the existing "supply chain" can structure itself non-cooperatively based on plant locations. The paper did not advise complete cooperation in the supply chain to generate better returns. Authors in [18] formulated and solved a problem of joint production planning, retailer selection and pricing for a manufacturer constrained by emissions' regulatory limits. Authors developed a mixed nonlinear bilevel programming problem where retailers' and manufacturer's profits were maximized. They solved the problem using a nested genetic algorithm, assuming the manufacturer as leader and retailers as followers in the multi-period Stackelberg game. They found that due to carbon emission regulations, a manufacturer will be obligated to select fewer (but the most profitable) retailers. Retailers, on the other hand, would increase the advertising expenditure and reduce their retail price if they want to be selected in the manufacturers' supply chain.

Many of these papers dealt with price competition between manufacturers and retailers when both players seek to maximize their respective profits. However, these papers were not tuned to deal with omnichannel retailing or situations in which multiple retail outlets are owned by both players. Our review suggests that there is a need to investigate aspects of price competition between manufacturer and retailer when the former operates a direct channel. Nevertheless, other aspects of the supply chain, such as production planning and inventory management, do exist as discussed in the previous paragraph.

Next, we discuss bilevel programming, a common theme that appeared in all these papers.

C. Bilevel Programming

Bilevel programming problems can be realized in several real-life scenarios, including supply chain coordination [14], toll optimization [19], the defense sector [20], production planning [21], facility location optimization [22], dynamic facility layout problems [23], road network design problems [24], bus rapid transit system design [25], and real-time pricing for smart grids [26].

Bilevel programming is employed to model decentralized management problem where two noncooperative decision-makers are in a hierarchical structure [16]. Bilevel optimization problems are essentially a special case of multilevel optimization and are very closely related to Stackelberg's leader– follower games [27]. The optimization involves solving two problems simultaneously, where one of the problems acts as a constraint to the other [28]. It contains two types of decision variables: a) upper-level (leader's) decision variables x where $x \in X \subseteq R^n$ and b) lower-level (follower's) decision variables y where $y \in Y \subseteq R^m$. In general, such problems are formulated as follows:

$$\max_{x \in X, y \in Y} F(x, y)$$

subject to:
$$y \in \underset{y \in Y}{\operatorname{arg\,max}} \left\{ f(x, y) : g_j(x, y) \ge 0, \ j = 1, ..., J \right\},$$
$$G_i(x, y) \ge 0, \ i = 1, ..., I$$

F(x, y) and f(x, y) are the leader's and the follower's objective functions, respectively. $G_i(x, y) \ge 0$ for i = 1, ..., I and $g_j(x, y) \ge 0$ for j = 1, ..., J are the constraints for the upper-level and the lower-level optimization problems, respectively. The literature suggests that these problems are mostly solved using one of two approaches: a) single-level reduction technique and b) hierarchical optimization technique. For a detailed review of the solution techniques, readers are requested to refer to [29].

D. Contributions

The paper makes the following contributions to the literature. First, it presents a joint production planning, pricing and inventory control problem for manufacturers operating in an omnichannel retailing

environment. Second, this paper formulates a bilevel price optimization problem where a make-to-stock manufacturer operates a direct online channel and competes with retailer's online and offline channels. Both the manufacturer and the retailer seek to maximize their respective profits before the season/inventory ends (assuming price and lead time-sensitive demand, cross-channel demand substitution and limited production capacity). Third, our model investigates the possibility of collaboration between the manufacturer and the retailer through a drop-shipping channel. Drop-shipping is considered as an omnichannel retailing service [30] where the retailer and the manufacturer share the profit earned through drop-shipping fulfillments. To solve the proposed problem, this paper evaluates several solution techniques including exact and heuristic approaches. Finally, based on the results obtained through numerical experiments, we generated some managerial insights for the manufacturer, specifically vis-à-vis drop-shipping channel.

The rest of the paper is structured as follows. In Section II, the problem scenario is described in greater detail; Section III presents the notations and formulations for the bilevel optimization problem. Section IV discusses and evaluates various solution techniques. Section V reports the numerical experiments and presents the results obtained using sensitivity analyses. Section VI discusses some of the managerial insights and outlines the scope for future research. Finally, Section VII concludes this paper.

II. PROBLEM DESCRIPTION

For this research, we consider that there is a make-to-stock manufacturer that is interested in maximizing the profit from the sale of its product. The manufacturer produces seasonal goods well in advance and sells them during the appropriate season. It can carry out production during the selling season, but production capacity during the selling season is constrained [15]. The manufacturer sells the products via a traditional retailer, i.e., a multi-location online-offline retailer like Walmart. Additionally, it also owns an online direct channel. In collaboration with the retailer, there is a provision to incorporate omnichannel retailing by providing the customers with additional drop-shipping alternative [30]. A customer can place an order directly to the manufacturer through any of the retailer-owned stores (online or offline). The manufacturer fulfills both direct-channel and drop-shipping orders with the help of a third-party logistics (3PL) provider. A vendor-managed inventory (VMI) model has been adopted as it enables the manufacturer to achieve better production planning by retrieving the downstream market information. VMI also helps

retailers improve their fill rate and decrease inventory stock-outs. Therefore, it is assumed that inventories at the retailer-owned stores are managed by the manufacturer.

This problem is further elaborated with the help of Fig. 1. In this figure, it is shown that the market contains two zones (although there can be any number of zones). A customer can place an order through any of the four available channels present in a zone. They can buy the product directly from the store, place a nonline order through the retailer's e-commerce website, or place a drop-shipping order to the manufacturer through either of the retailer's store or online channels. When an order is routed through either of the retailer receives a share of the profit made by the manufacturer. A customer can also order the product through the manufacturer's direct channel, thus bypassing the retailer altogether. If the retailer receives an online order from a customer located in zone *z*, the product is delivered within a day using inventory available in the store located in zone *z*. However, if an order is placed to the manufacturer through either drop-shipping or direct channel, it is assumed that the product will be delivered after a certain delivery lead time. "Delivery lead time" is the time gap between the placement of an online order and the actual delivery of the product. In both drop-shipping and direct channel, delivery lead time can influence overall demand; therefore, demand is considered to be price- and time-sensitive. Nevertheless, it is assumed that the manufacturer can expedite the deliveries; however, expediting a delivery increases the transportation cost and decreases the profit margin for the manufacturer.

In summary, the problem considers that the manufacturer controls the wholesale price, the directchannel and the drop-shipping prices, the production quantity, inventory at the retailer's stores across all the zones, and fulfillments through direct and drop-shipping channels. Additionally, the manufacturer can decide when to expedite the fulfillments of direct and drop-shipping orders. However, in order to improve the delivery lead times, the manufacturer must pay a relatively higher price to the 3PL partner.



Fig. 1: Flow of information and the product in omnichannel retailing considered in this paper.

The retailer determines the online and offline prices, as well as the fulfillments through respective channels. As a business rule, at a given point in time, the retailer's online channel should offer a uniform price irrespective of the zone. But there can be a price difference between individual stores and online channel. Similarly, the manufacturer follows price-matching in the direct and drop-shipping channels at a given point in time. The wholesale price and retailer's share in the profit generated from the drop-shipping channel is fixed at the start of the selling season and remains so over the entire season.

III. MODEL FORMULATION

A. Notations

Sets:

Z: all market zones

T: all selling periods

 Ω : the set of all prices

Indices:

8

z: index for a market zone, $z \in Z$

t: index for periods, $t \in T$

Input Parameters:

 c_m : per-unit manufacturing cost of the product

 M_z^t : size of the market in market zone z and time t

 c_z : transportation cost for a unit of product from manufacturer to retailer's store in market zone z c_z^f : fixed cost of replenishments of the retailer's store in market zone z, borne by the manufacturer h_z : holding cost per unit time per unit product in the store located in market zone z h_m : holding cost per unit time per unit product in the manufacturer's warehouse c_{lz} : price charged by the 3PL partner for delivery to a customer in zone z c_{oz} : least incurred fulfillment costs when the delivery lead time is very long for zone z γ_z : constant representing the increase in cost for expediting delivery of an order c_{erz} : cost of online demand fulfillment from the store to a customer in market zone z α_{bz} : intrinsic attraction towards the product in retailer's store channel in zone z β_{bz}^{p} : price sensitivity towards the product in retailer's store channel in zone z α_{erz} : intrinsic attraction towards the product in retailer's online channel in zone z β_{erz}^{p} : price sensitivity towards the product in retailer's online channel in zone z α_{emz} : intrinsic attraction towards the product in manufacturer's direct channel in zone z β_{emz}^p : price sensitivity towards the product in manufacturer's direct channel in zone z β_{emz}^{l} : delivery lead-time sensitivity towards the product in manufacturer's direct channel in zone z α_{bmz} : intrinsic attraction towards the product in the drop-shipping channel in zone z β_{hmz}^p : price sensitivity towards the product in the drop-shipping channel in zone z β_{bmz}^{l} : delivery lead-time sensitivity towards the product in the drop-shipping channel in zone z Discretionary decision variables:

 q_m : represents allocated production capacity per unit time t

W: wholesale price set by the manufacturer at the start of the selling season

 θ : retailer's share in the profit earned through the drop-shipping demand fulfillments

Upper-level decision variables:

 q_m^t : production during the period t where $q_m^t \le q_m$

 B_z^t : binary decision variable for the replenishment of the retailer's store in zone *z* at time *t* l_z^t : delivery lead time for demand fulfillments in market zone *z* and time *t* S_{emz}^t : total direct channel demand fulfillments by the manufacturer in zone *z* and time *t* S_{bmz}^t : total drop-shipping demand fulfillments by the manufacturer in zone *z* and time *t* Q_z^t : replenishment quantities to the retailer's store in zone *z* and time *t* I_z^t : level of inventory in the retailer's store in zone *z* and at the end of time *t* I_z^n : level of inventory in the manufacturer's warehouse at the end of time *t* I_z^o : initial inventory made available in the retailer's store in zone *z* before sales start I_m^o : initial inventory available in the manufacturer's warehouse at the start of the season P_{em}^t : the price of the product in direct and drop-shipping channels at time *t Lower-level decision variables:*

 P_{bz}^{t} : price listed by the retailer in its store located in market zone z at time t P_{er}^{t} : price listed by the retailer on its online channel at time t S_{bz}^{t} : total store channel demand fulfilled by the retailer, in zone z and time t

 S_{erz}^t : total online channel demand fulfilled by the retailer, in zone z and time t

B. Demand Estimation

In the presence of multiple channels and an almost negligible hassle of alternating between available channels, an increase in the price in one of the channels can decrease its share from the total sales. This decrease in sales often gets distributed among other channels; this phenomenon is known as "cross-channel demand substitution" [12]. In the past, researchers have adopted a variety of mathematical models to understand demand substitution; for a comprehensive review, please refer to [31]. We note that discrete choice models are some of the commonly used demand-modeling techniques [32]. Although cross-channel demand substitution can also be modeled using linear demand functions, using discrete choice models over the former yields the benefit of parsimony of the number of coefficients that must be estimated [33].

Whereas discrete choice models need O(|M|) coefficients, linear demand functions may require $O(|M|^2)$ coefficients [12].

Discrete choice models like multinomial logit (MNL) and multiplicative competitive interaction (MCI) are extensively used in fields like transportation planning [34], assortment optimization [35], and marketing research [36]. They help model customer behavior where many substitutable alternatives are available. These models are based on the assumption that customers associate a utility with each available alternative, including a zero utility for not choosing any alternative [37]. Using a discrete choice model, demand in channel $n, n \in N$ and at location $z, z \in Z$ can be estimated using the following equation:

$$D_{nz} = M_z * \frac{A_{nz}}{1 + \sum_{n' \in N} A_{n'z}}$$
(1)

where M_z represents the size of the market; A_{nz} (also known as an attraction function) is a function of explanatory variables such as price, lead time and/or advertisement, and helps determine demand generated in channel *n* at location *z*. In this paper, we have employed an MNL discrete choice model to estimate the

demand. In MNL models, attraction functions are formulated as $A_{nz} = \exp\left(\alpha_{nz} + \sum_{k=1}^{K} \beta_{nz}^{k} k_{nz}\right)$, where K

represents total number of explanatory variables such as price, delivery lead time and quality; α_{nz} represents the intrinsic attraction constant of the product in channel *n*; β_{nz}^{k} represents the sensitivity of the customers toward the explanatory variable *k*. For a detailed discussion on techniques used to estimate these demand parameters, please refer to [32], [38].

C. Assumptions

Assumption 1: Both the manufacturer and the retailer are interested in profit maximization.

Assumption 2: The manufacturer is responsible for managing the retailer's inventory under the VMI strategy. This enables the manufacturer to have downstream market information.

Assumption 3: Demand is price- and lead-time-sensitive [39], and depending on the degree of product differentiation [7], attraction towards retailer's offline, retailer's online, manufacturer's direct and drop-shipping channels can be modeled as follows:

 $A_{bz}^{t} = \exp(\alpha_{bz} - \beta_{bz}^{p} P_{bz}^{t})$, where A_{bz}^{t} represents the attraction towards the product available in

retailer's physical store present in zone z and time $t; A_{erz}^{t} = \exp(\alpha_{erz} - \beta_{erz}^{p} P_{er}^{t})$, where A_{erz}^{t} represents the attraction towards the product in retailer's online channel in zone z and time t;

 $A_{emz}^{t} = \exp\left(\alpha_{emz} - \beta_{emz}^{p} P_{em}^{t} - \beta_{emz}^{l} l_{emz}^{t}\right), \text{ where } A_{emz}^{t} \text{ represents the attraction towards the product in}$ manufacturer's direct channel in zone z and time t; and, finally, $A_{bmz}^{t} = \exp\left(\alpha_{bmz} - \beta_{bmz}^{p} P_{em}^{t} - \beta_{bmz}^{l} l_{bmz}^{t}\right),$ where A_{bmz}^{t} signifies the attraction towards the product available via the drop-shipping channel in zone z and time t.

Assumption 4: The manufacturer can expedite direct and drop-shipping deliveries with the help of a 3PL partner. Cost of expediting deliveries can be calculated using $c_{lz} = c_{oz} + \gamma_z/l_z$ [39], where c_{lz} represents the delivery cost, l_z denotes the delivery lead time, γ_z signifies the marginal increase in the cost for expediting the deliveries, and c_{oz} represents the bare minimum fulfillment cost for zone z.

Assumption 5: Retailer's online orders are fulfilled from the nearest physical store, and because all the deliveries are local, a fixed cost is incurred for every such delivery.

Assumption 6: Retailer's online channel offers uniform price irrespective of the customer's location.

Similarly, manufacturer's direct channel and drop-shipping channel offers uniform price irrespective of the customer's location.

D. Objective Functions and Constraints

Manufacturer's Profit Function (Upper-Level Problem):

Maximize:

$$\sum_{z=1}^{Z} \sum_{t=1}^{T} (W - c_z) B_z^t Q_z^t + \sum_{t=1}^{T} \sum_{z=1}^{Z} (P_{em}^t - c_{lz}^t - \frac{h_m}{2}) S_{emz}^t + \sum_{t=1}^{T} \sum_{z=1}^{Z} (P_{em}^t - c_m - c_{lz}^t - \frac{h_m}{2}) (1 - \theta) S_{bmz}^t - \sum_{t=1}^{T} \sum_{z=1}^{Z} c_z^f B_z^t - \sum_{t=1}^{T} I_m^t h_m - I_m^t \frac{h_m I_m^t}{2q_m} - c_m \left(\sum_{t=1}^{T} q_m^t - \sum_{t=1}^{T} \sum_{z=1}^{Z} S_{bmz}^t\right) - \sum_{z=1}^{Z} I_z^T W$$
(2)

Subject to:

$$S_{emz}^{t} \leq M_{z}^{t} \left(\frac{A_{emz}^{t}}{1 + A_{bz}^{t} + A_{emz}^{t} + A_{bmz}^{t}} \right) \qquad \forall z \in \mathbb{Z}, \forall t \in \mathbb{T}$$

$$(3)$$

$$S_{bmz}^{t} \leq M_{z}^{t} \left(\frac{A_{bmz}^{t}}{1 + A_{bz}^{t} + A_{erz}^{t} + A_{bmz}^{t}} \right) \qquad \forall z \in \mathbb{Z}, \forall t \in \mathbb{T}$$

$$(4)$$

$$Q_{z}^{t}, S_{bmz}^{t}, S_{emz}^{t} \ge 0 \qquad \qquad \forall z \in \mathbb{Z}, \ \forall t \in \mathbb{I}$$

$$(5)$$

$$I_{m}^{t}, I_{z}^{t} > 0 \qquad \qquad \forall z \in \mathbb{Z}, \, \forall t \in \mathbb{T}$$

$$\tag{6}$$

$$q_m^t \le q_m \qquad \qquad \forall t \in T \tag{7}$$

$$B_{z}^{t} \in \{0,1\} \qquad \forall z \in \mathbb{Z}, \, \forall t \in \mathbb{T}$$

$$(8)$$

$$p'_{em} \in \Omega \qquad \qquad \forall z \in Z, \ \forall t \in T \tag{9}$$

Retailer's Profit Function (Lower-Level Problem):

Maximize: $\sum_{t=1}^{T} \sum_{z=1}^{Z} \left(P_{bz}^{t} - W - \frac{h_{z}}{2} \right) S_{bz}^{t} + \sum_{t=1}^{T} \sum_{z=1}^{Z} \left(P_{er}^{t} - W - \frac{h_{z}}{2} \right) S_{erz}^{t} + \sum_{t=1}^{T} \sum_{z=1}^{Z} \left(P_{em}^{t} - c_{m} - c_{lz}^{t} - \frac{h_{m}}{2} \right) \theta S_{bmz}^{t} - \sum_{z=1}^{Z} \sum_{t=1}^{T} c_{erz} S_{erz}^{t} - \sum_{t=1}^{T} \sum_{z=1}^{Z} I_{z}^{t} h_{z}$ (10)

Subject to:

$$S_{bz}^{t} \leq M_{z}^{t} \left(\frac{A_{bz}^{t}}{1 + A_{bz}^{t} + A_{enz}^{t} + A_{bmz}^{t}} \right) \qquad \forall z \in \mathbb{Z}, \forall t \in \mathbb{T}$$

$$(11)$$

$$S_{erz}^{t} \leq M_{z}^{t} \left(\frac{A_{erz}^{t}}{1 + A_{bz}^{t} + A_{erz}^{t} + A_{emz}^{t} + A_{bmz}^{t}} \right) \qquad \forall z \in \mathbb{Z}, \forall t \in \mathbb{T}$$

$$(12)$$

$$S_{bz}^{t}, S_{erz}^{t} \ge 0 \qquad \qquad \forall z \in \mathbb{Z}, \ \forall t \in \mathbb{T}$$
(13)

$$p_{er}^{t}, p_{bz}^{t} \in \Omega \qquad \qquad \forall z \in Z, \, \forall t \in T$$
(14)

Equation (2) represents the manufacturer's profit in the proposed bilevel omnichannel price optimization problem. The first term in the manufacturer's objective function is the profit earned by the manufacturer by supplying the product to the retailer at a wholesale price W. The second term is the profit earned by the manufacturer through direct-channel demand fulfillments. The third term is the manufacturer's share in the profit earned through drop-shipping demand fulfillments. It has been assumed that W and θ are determined by the manufacturer at the start of the selling season and remain constant for the entire season. The fourth term represents the total replenishment costs incurred by the manufacturer to replenish the retailer-owned stores. Inventory holding costs at the manufacturer's warehouse are represented by the next two terms in the objective function. The seventh term is the manufacturing cost incurred to the manufacturer. Since the manufacturing cost is already removed from the drop-shipping profits (see the third term), S_{bmz}^t for all $t, t \in T$ and all $z, z \in Z$ is reduced from the total quantity produced over the time $t, t \in T$. Finally, the last term in the manufacturer's objective function represents recoveries of unsold items from the retailer's stores at the end of the selling period t, where t = T.

Equations (3) and (4) are the constraints so that the fulfillments cannot be more than the direct and the drop-shipping demands, respectively. Constraints in (5) are non-negativity constraints and represent that production or fulfillments during an interval cannot be less than zero. Constraints in (6) signify non-negativity of the manufacturer and the retailer's inventories, where I_m^t and I_z^t are calculated using (15) and (16), respectively.

$$I_{m}^{t} = I_{m}^{o} + \sum_{n=1}^{t} q_{m}^{n} - \sum_{n=1}^{t} \sum_{z=1}^{Z} \left(B_{z}^{n} Q_{z}^{n} + S_{emz}^{n} + S_{bmz}^{n} \right) \qquad \forall z \in Z, \ \forall t \in T$$

$$I_{z}^{t} = I_{z}^{o} + \sum_{n=1}^{t} \left(B_{z}^{n} Q_{z}^{n} - S_{bz}^{n} - S_{erz}^{n} \right) \qquad \forall z \in Z, \ \forall t \in T$$
(15)

Constraint (7) imposes a limit on variation in the units produced per unit time considering allocated production capacity. We know that once the resources are assigned to a certain task, it is difficult to reallocate them to some other activities. Companies often use overtime when actual demand is high, but there is a limit to which the productivity of the resources can be extended. Reducing the volume of production is relatively easy, but that means keeping the resources under-utilized. In a way, q_m^t is a variable that is difficult to adjust once the resources are allocated for q_m units. Constraint (8) represents the binary decision variable B_z^t ; if $B_z^t = 1$, the manufacturer replenishes the retailer's store present in zone z at time t with a quantity Q_z^t where $Q_z^t \ge 0$. Constraint (9) ensures that direct and drop-shipping prices belong to a set of predefined prices Ω .

Equation (10) defines the retailer's profit; the first two terms in the formulation represent the profit from the stores and the online channel, respectively. The third term represents the retailer's share in the profit from the drop-shipping fulfillments. One can see that the profit earned by the retailer through the drop-shipping channel depends on the manufacturer's decision variable θ . The fourth term is the total cost incurred by the retailer to fulfill the online orders. Finally, there is an inventory holding cost involved. Constraints (11) and (12) ensure that the fulfillments are not more than the demand in the store or the online channel, respectively. Equation (13) represents the non-negativity of fulfillments. Constraint in (14) ensures that the store and the online prices belong to a set of predefined prices Ω .

IV. SOLUTION APPROACH

Bilevel programming problems contain hierarchical optimization structures and are challenging to solve. Even if the objective functions of both players are linear, the overall problem remains NP-Hard [40]. Often, bilevel programming problems are solved using a single-level reduction technique, which uses the classical KKT approach while assuming the smoothness, linearity, or convexity of the objective functions. But the real-world bilevel optimization problem can contain non-differentiability and non-convexity [41]. This reduces our ability to convert the bilevel programming problems to a classic single-level optimization problem. These non-linearities force researchers to opt for heuristic and evolutionary approaches to solve the problem; for example, see [14], [16]–[18], [23].

It is known that MNL profit functions are not convex [42]. Therefore, employing KKT conditions directly on the retailer's problem for single-level reduction will not guarantee an optimal solution. Nonetheless, we can use single-level reduction using KKT conditions if we can convexify the retailer's problem.

Proposition 1: In the single-location multi-period bilevel price optimization problem, the retailer's objective function when reformulated to purchase probability space $(\lambda_{er}, \lambda_b)$ instead of price (P_{er}, P_b) is a concave function, given that all the demand generated at the retailer's stores is fulfilled.

Proof: See Appendix

This proposition states that if we consider a retailer that operates only one physical store and ensures that all the demand that gets generated is fulfilled, then the retailer's problem can be convexified. Convexification allows us to use the single-level reduction technique and thereby solve single-location bilevel optimization problems. As a result, the problem is now divided into two cases: Case 1, the single-location, multi-period bilevel omnichannel price optimization problem, and Case 2, the multi-location, multi-period bilevel omnichannel price optimization problem.

Case 1: Single-Location, Multi-Period Bilevel Omnichannel Price Optimization Problem

Based on Proposition 1, the retailer's single location problem (Eqs. 10–14) can be replaced by the objective function show in (17), with no constraints to satisfy and assuming holding costs are negligible (for proof see (A.6) in the Appendix).

Maximize:
$$\Pi_{r} = \left(\frac{1}{\beta_{b}^{p}}\left(\alpha_{b}^{p} - \ln\left(\frac{\lambda_{b}}{1 - \lambda_{b} - \lambda_{er} - \lambda_{em} - \lambda_{bm}}\right)\right) - W\right)M\lambda_{b} + \left(\frac{1}{\beta_{er}^{p}}\left(\alpha_{er}^{p} - \ln\left(\frac{\lambda_{er}}{1 - \lambda_{b} - \lambda_{er} - \lambda_{em} - \lambda_{bm}}\right)\right) - W\right)M\lambda_{er} + \left(\frac{1}{\beta_{bm}^{p}}\left(\alpha_{bm}^{p} - \ln\left(\frac{\lambda_{bm}}{1 - \lambda_{b} - \lambda_{er} - \lambda_{em} - \lambda_{bm}}\right) - \beta_{bm}^{l}I_{bm}\right) - c_{m} - c_{l}\right)\theta M\lambda_{bm} - c_{er}M\lambda_{er}$$

$$(17)$$

Furthermore, since (17) is concave in purchase probability space λ_{er} and λ_b , it can be replaced by its KKT conditions (18) and (19), where λ_{er} and λ_b are retailer's alternate decision variables.

$$\frac{\partial \Pi_{r}}{\partial \lambda_{\rm er}} = \frac{\theta M \lambda_{\rm bm}}{\beta_{\rm bm}(-1+\lambda_{\rm bm}+\lambda_{\rm b}+\lambda_{\rm em}+\lambda_{\rm er})} + \frac{M \lambda_{\rm b}}{\beta_{\rm bz}(-1+\lambda_{\rm bm}+\lambda_{\rm b}+\lambda_{\rm em}+\lambda_{\rm er})} - \frac{M(-1+\lambda_{\rm bm}+\lambda_{\rm b}+\lambda_{\rm em})}{\beta_{\rm er}(-1+\lambda_{\rm bm}+\lambda_{\rm b}+\lambda_{\rm em}+\lambda_{\rm er})} - \frac{M}{\beta_{\rm er}(-1+\lambda_{\rm bm}+\lambda_{\rm b}+\lambda_{\rm em}+\lambda_{\rm er})}{-1+\lambda_{\rm bm}+\lambda_{\rm b}+\lambda_{\rm em}+\lambda_{\rm er}} = 0$$

$$\frac{\partial \Pi_{r}}{\partial \Pi_{r}} = \frac{\theta M \lambda_{\rm bm}}{\theta M \lambda_{\rm bm}} + \frac{M \lambda_{\rm er}}{1+\lambda_{\rm bm}+\lambda_{\rm em}+\lambda_{\rm er}} = 0$$

$$(18)$$

$$\frac{\partial \lambda_{b}}{\partial \lambda_{b}} = \frac{\partial M}{\beta_{bm}(-1+\lambda_{bm}+\lambda_{b}+\lambda_{em}+\lambda_{er})} + \frac{\partial M}{\beta_{er}(-1+\lambda_{bm}+\lambda_{b}+\lambda_{em}+\lambda_{er})} - \frac{\partial M}{\beta_{b}(-1+\lambda_{bm}+\lambda_{b}+\lambda_{em}+\lambda_{er})} - \frac{\partial M}{\beta_{b}(-1+\lambda_{bm}+\lambda_{b}+\lambda_{em}+\lambda_{em})} -$$

It must be noted that, although we were able to simplify the bilevel problem to a single-level problem, the manufacturer's objective function remains a non-convex NP-Hard problem. Therefore, the manufacturer's problem, along with the newly added constraints from the retailer's problem (Eqs. 18–19) must be solved using stochastic search algorithms. We opt for Simulated Annealing (SA) [43] and Randomized Decomposition Solver (RDSolver)[44] because of the discrete nature of the decision variables we are dealing with.

Case 2: Multi-Location, Multi-Period Bilevel Omnichannel Price Optimization Problem

Since the retailer's online price across multiple locations is supposed to be equal, the retailer's problem cannot be convexified for the multi-location problem (refer to Proposition 2 in [12]). Therefore, in this case, the upper- and lower-level problems must be solved iteratively and cannot be simplified using the single-level reduction technique as in Case 1. This makes it computationally more challenging as compared to solving a single-level NP-Hard problem. Subsequently, to solve the proposed multi-location, multi-period bilevel price optimization problem, we employed a frequently used technique known as hierarchical bilevel optimization [16], [45].



A. Hierarchical bilevel optimization

Fig. 2: The flowchart of hierarchical bilevel optimization.

Fig. 2 presents the flowchart of hierarchical bilevel optimization. In step 1, a random solution for each player is generated. Let x represent a random solution for the manufacturer and y represent a random

solution for the retailer. In step 2, keeping the manufacturer's decision variables x as constants, the retailer's problem is solved completely. The optimal response y^* obtained for the retailer in step 2, along with the previously found solution x, is then forwarded to step 3. In step 3, the manufacturer's problem is solved completely, keeping the retailer's decision variables y^* as constants. The obtained solutions (x^*, y^*) are then sent to step 4. In step 4, if any of the players' objective function improves because of (x^*, y^*) , we accept the solution and assign $x = x^*$ and $y = y^*$. Finally, in step 5 we check whether the algorithm has converged. It is presumed that the algorithm has converged if there is no change in either x or y for consecutive n number of iterations of steps 2 to 4. If the termination criterion is not satisfied, the flow returns to step 2 where the process starts with the updated values of x and y. To carry out optimization in steps 2 and 3, we once again used Simulated Annealing (SA) [43] and Randomized Decomposition Solver (RDSolver) [44] because of their ability to handle discrete decision variables. For more details on these optimization techniques, please see the next section.

B. SA and RDSolver

SA is a well-known algorithm originally proposed by [43]. It was inspired by the process of annealing in metals, in which a metal is heated to a temperature high enough for its molecular structure to break down. Subsequently, the metal is slowly cooled to obtain a desired molecular structure. SA is often considered to be a discrete optimization algorithm and therefore fits well into the requirements of our problem. Moreover, SA has been used by many researchers in the past to solve bilevel optimization problems; for an example, see [46].

RDSolver is a solution framework recently proposed by [44]. It was developed to solve nonlinear, nonconvex discrete optimization problems by combining two algorithms, namely Randomized Decomposition (RD) and RDPerturb. RD is a local search approach that starts from a feasible solution and is based on splitting the decision variables into random subsets. To escape local optima, RDPerturb is used along with RD. In the original paper, RDSolver was tested on 400 problem instances and the results were found to be comparable with that of the state-of-the-art solution techniques. As RDSolver needs no prior knowledge of the structure of the problem and can solve discrete optimization problems, the algorithm fits

well with our requirements. For pseudo-codes for RDSolver, RD and RDPerturb, readers are invited to refer to algorithms 1, 2, and 3 in [44].

C. Comparative Analysis

Since the algorithms under consideration are known to provide near-optimal results, we must assess the quality of the solutions obtained. In Case 1, we discussed how to convert our single-zone bilevel problem into a single-level problem where only one objective function must be optimized. Because of its straightforward application, solutions obtained after single-level reduction (as in Case 1) can also serve as a benchmark for hierarchical bilevel optimization techniques (intended for multi-location problems). Hence, we carried out a comparative analysis to see how the objective functions converged, and the best/worst solutions obtained.

For benchmarking and comparative analysis, a problem instance consisting of one zone and three period was created. Parameter tuning was carried out for all the given algorithms. The problem instance was then solved multiple times using each of the algorithms. The results were recorded for statistical analysis. Figs. 3a and 3b present how the results converged when we used SA, and the problem was reduced to single-level versus when the problem was solved hierarchically for upper and lower levels separately. Figs. 3c and 3d present the convergence when RDSolver was used, and the problem was reduced to single-level versus when the problem was solved hierarchically for upper and lower levels separately. The convergence signifies that neither the manufacturer nor the retailer wants to deviate from the near-equilibrium attained, also known as "Stackelberg equilibrium."

In the one-zone three-period problem scenario, these algorithms took approximately five to ten minutes to converge. Since the algorithms are based on stochastic search techniques, the time taken by them to converge may vary. The speed of convergence also depends on how the algorithmic parameters are tuned. However, we observe that, on an average, when the problem was reduced to single level, the time to convergence was less than when the problem was solved using the hierarchical bilevel optimization route. This is because it is always straightforward and simple to solve a traditional single-level optimization problem than to solve a bilevel problem.

Table 1 shows the compiled results of mean, standard deviation, best, and worst values obtained using all the mentioned solution techniques. On average, the retailer's profit converged at approximately 200, whereas the manufacturer's profit converged at approximately 2000. Based on these results, it can be said that the single-level reduction technique provided marginally inferior optimal values as compared to the hierarchical optimization technique. The reason may be that for single-level reduction, we have simplified the original problem by relaxing the constraints (see Appendix).

Since the results obtained using all four solution techniques are almost identical, we can safely proceed to solve multi-location problems using the hierarchical bilevel optimization technique as shown in Fig. 2. Furthermore, as shown in Table 1, since hierarchical-RDSolver gave relatively better results, it may be preferred over hierarchical-SA.



Fig. 3a: Convergence of manufacturer's and retailer's profit when the problem was reduced to single-level and solved using SA.



Fig. 3c: Convergence of manufacturer's and retailer's profit when the problem was reduced to single-level and solved using RDSolver.



Fig. 3b: Convergence of manufacturer's and retailer's profit when the problem was solved using hierarchical bilevel optimization using SA.



Fig. 3d: Convergence of manufacturer's and retailer's profit when the problem was solved using hierarchical bilevel optimization using RDSolver.

	SA on Single-Level Reduced Problem		SA for Bilevel		RDSolver on Single-		RDSolver for Bilevel	
			Optimization		Level Problem		Optimization	
Player	Mfr.	Ret.	Mfr.	Ret.	Mfr.	Ret.	Mfr.	Ret.
Mean	1989.14	131.56	2022.56	169.64	1932.83	207.84	2242.12	201.68
Std. Dev.	124.98	48.52	136.79	41.06	65.72	51.95	61.67	48.65
Best	2120	182	2281.4	237.6	2011	213	2323	268
Worst	1796.3	118.2	1877.4	142.6	1817	287	2155	168

Table 1: Mean, standard deviation, best and worst values obtained using various solution techniques.

In the following section, we have compared hierarchical-SA and hierarchical-RDSolver based on the time they take to converge while solving different problem instances. Fig. 4a presents the number of decision variables for respective problem instances. Fig. 4b presents box plots for the time taken by these algorithms to converge. We want to highlight that all the experiments were done on an Intel i5-6500 with a 3.20 GHz processor and 8 GB RAM.

900

800





decision variables.





optimization.

The results presented in Fig. 4b indicates an exponential growth in computational time as problem size increases. Hierarchical bilevel optimization based on RDSolver took relatively longer to converge (often more than 10 hours). Our analysis suggests that the reason for the slow convergence of larger problem instances relates to the nature of the problem itself. As we are solving a game-theory problem, for every value-combination of the manufacturer's decision-variables, the retailer has a corresponding optimal response.

Because both the players' objective functions are non-convex and non-differentiable, finding the retailer's optimal response that corresponds to that of the manufacturer's existing solution is itself computationally challenging. The algorithm will converge only when neither of the players has a better solution to counter the existing solution of the other player. Furthermore, every time either of the player's decision variables changes, the other player's solution must also be optimized to counter the change. In bilevel optimization, search space increases exponentially for every new decision variable added to either of the players' objective function [29]. This is because the algorithm is not optimizing both the functions concurrently (rather, it optimizes sequentially). That is the why we sought to reduce this bilevel problem to a single-level problem. However, we found that multi-location problems are difficult to simplify.

V. NUMERICAL EXPERIMENTS

This section reports the results obtained using the numerical experiments. Data and model parameters were adopted from the literature. Various scenarios concerning market variables, such as intrinsic utilities, price sensitivities and delivery lead-time sensitivity, were investigated. It is assumed that there are three zones ($z \in [1,3]$) and that the selling season is divided into five periods ($t \in [1,5]$). The approximate market size (M_z^t) is assumed to be known and is presented in Fig. 5.



Fig. 5: Market size for the product for various zones and times.

The cost of manufacturing a unit of product was fixed and is assumed to be 110, i.e., $c_m=110$. To replenish the retailer's store located in zone $z \in [1,3]$, c_z^f was assumed to be 70, 120 and 150, respectively. An additional cost per unit transported from the manufacturer's warehouse to the retailer's store (c_z) was assumed to be 1. Holding costs per period (h_z) for $z \in [1,3]$ were assumed to be 3, 2.5 and 2, respectively. Holding cost at the manufacturer's facility (h_m) was assumed to be 1 per unit time. For direct and dropshipping fulfillments, extra cost for expediting deliveries was calculated using γ_z/l_z . Here γ_z for $z \in [1,3]$ were assumed to be 2, 4 and 6, respectively. The minimum amounts charged by 3PL for a delivery to a customer located in zone $z \in [1,3]$, i.e., c_{oz} were assumed to be 1, 2 and 3 per unit. An online demand fulfillment cost for the retailer (c_{erz}) was assumed to be 1 per unit. It was assumed to be minimal because the deliveries are performed locally from the store located in that zone.

Next, it was assumed that $\alpha_{bz}=124$, $\alpha_{erz}=126$, $\alpha_{bmz}=126$ and $\alpha_{emz}=124$ were the intrinsic utilities for retailer's stores and online channel, drop-shipping channel, and manufacturer's direct channel, respectively. Furthermore, $\beta_{bz}^p=0.88$, $\beta_{erz}^p=0.9$, $\beta_{emz}^p=0.9$ and $\beta_{bmz}^p=0.9$ were the assumed price sensitivities in the retailer's stores and online channel, drop-shipping channel, and manufacturer's direct channel, respectively. Finally, delivery lead-time sensitivities β_{emz}^l and β_{bmz}^l were assumed to be 0.25, whereas delivery lead times l_{emz}^t and l_{bmz}^t varied from 1 to 5 days. Later we increased these sensitivities towards 0.50 and 0.75 to evaluate their effect on profitabilities.

In the manufacturer's search space, there are discretionary variables and operational variables. Variables those are classified as discretionary (or strategic level) are W, θ , and q_m . These variables have a significant impact on other decision variables and ultimately on the convergence of the algorithms. Every time any of these three variables were modified, the prior solution (near-equilibrium) obtained became obsolete, and the algorithm took longer to converge. Therefore, we decomposed the problem by eliminating W and θ from the search space, although allowed q_m to be optimized by the algorithm. However, doing so we had to optimize for various values of W and θ to find a combination that suits the manufacturer. Initial inventory (I_m^o) depends on the number of periods the production of q_m units start in advance. It was assumed that the manufacturer begins the production five interval before the start of the selling season such that $I_m^o = 5q_m$. Keeping the problem simple we also assumed $q_m^t = q_m$.

A. Finding Optimal Combination of W and θ

First, we solved the given problem by removing drop-shipping as a viable retail alternative and $\theta = 0$. Fig. 6a presents the profit earned by the players and total supply chain profit as *W*, which varied from 131 to 139. We can see that the retailer's profit decreases almost linearly as *W* increases. At the same time,

the manufacturer's profit increased to W = 135 and then started decreasing. At W = 135, the profit earned by the manufacturer is approximately 7450, while the retailer's profit is approximately 1350. The total supply chain profit stands at ≈ 8800 . These values are treated as benchmarks to evaluate the performance of the supply chain when drop-shipping is incorporated.

Now considering that the players are willing to collaborate over drop-shipping channel, we varied θ from 10% to 50%. In Fig. 6b we present how the manufacturer's profit decreased as θ increased. Similarly, in Fig. 6c we present how the retailer's profit increased as θ increased. And finally, in Fig. 6d the change in total supply chain profit is shown.



Fig. 6a: Change in supply-chain profit with respect





Fig. 6b: Change in manufacturer's profit with

respect to θ for given W.



Fig. 6c: Change in retailer's profit with respect to Fig. 6d: Change in supply-chain profit with respect

θ for given W.

to θ for given W.

In these figures (Figs. 6b–6d), we can that see when $\theta < 15\%$, the retailer's profit falls below the benchmark profit of 1350; therefore, in such a case, the retailer will not collaborate with the manufacturer. The two players will then operate in a competitive mode, and the supply-chain profit will drop to 8800.

When $\theta < 20\%$, the retailer reduces online and offline prices to compete with the now relatively less profitable drop-shipping channel. Although increasing θ from 20% to 30% seems to be counterproductive for the manufacturer, the change helps in terms of maximizing the overall profit. Total supply-chain profit increases to $\theta = 30\%$ but then decreases gradually. When $\theta \ge 40\%$, we see a lesser number of dropshipping fulfillments. The retailer's profit does not increase significantly after $\theta = 40\%$. We also observed a slight increase in the direct and drop-shipping prices. This is because having a drop-shipping channel becomes less profitable for the manufacturer when a large portion of the channel's profit goes to the retailer. When θ is more than 45%, the manufacturer's profit falls below the benchmark profit of 7450, rendering drop-shipping a non-viable solution once again.

For the manufacturer, it is optimal to select $\theta \approx 20\%$ where the retailer also has some incentive to collaborate. At this point, the overall fulfillment through drop-shipping was recorded to be the highest. However, the total supply-chain profit was maximized at $\theta = 30\%$. Our analysis suggests that: a) there is a scope for coordination between the two players when it comes to determining the value for θ , and b) the inclusion of drop-shipping channel in the channel mix can not only serve as an omnichannel feature but can also assist in generating higher returns, given the value of θ is selected judiciously.

B. Constrained Production Capacity

In the previous section, it was assumed that the manufacturer first determines the values of W and θ , and accordingly allocates the resources for production. It was found that for W = 135 and $\theta = 20\%$, $q_m = 50$ represents optimal production quantity per unit time. This value of q_m corresponds to approximately 82% of the market potential being satisfied. But what if the manufacturer is dealing with underproduction and overproduction because of its inability to allocate and reallocate resources for the same? In consideration of this question, to simulate over-production we assumed $q_m = 55$, and to simulate under-production we assumed $q_m = 40$ and solved for W and θ .

Case A: Under-production

When q_m was fixed at 40 units, optimal values of W and θ changed to 137 and 10%, respectively. For $q_m = 40$, it became difficult for the manufacturer to meet the retailer's demand while keeping the wholesale price at W = 135. W < 135 became infeasible because in VMI settings the supplier can be penalized for not fulfilling the retailer's demand [47].

Fig. 7a presents how drop-shipping affects the profitabilities in the supply chain in the case of under-production. At $\theta = 10\%$, drop-shipping helped increase the manufacturer's profit by up to 14% and that of the retailer by 4%. While at $\theta = 20\%$, the manufacturer's profit was 12% higher, the retailer's profit was 27% higher, and the total supply-chain profit was maximized.





Fig. 7a: Percentage change in profits with respect of θ for given W in the case of under-production.

Fig. 7b: Percentage change in profits with respect of θ for given W in the case of over-production.

Case B: Over-production

When q_m was fixed at 55 units, the optimal value for W changed to 133. In this case, since the manufacturer is under pressure to dump a larger volume of the product at lower W, it is interesting to see how the retailer perceived the drop-shipping channel.

Fig. 7b presents how drop-shipping affects the profitabilities in the supply chain in the case of overproduction. At $\theta = 10\%$, the retailer's profit increased by only 1%. At $\theta = 20\%$, where the retailer's profit is relatively high (increased 7%), the increase in the manufacturer's profit plummets back to just 1%. Since supporting a drop-shipping channel does not substantially increase the retailer's profit, the retailer may not find drop-shipping a game-changing proposition in the case of overproduction.

From the above analysis, we can infer that in the case that production capacity is not sufficient, a manufacturer can obtain better returns with the help of a drop-shipping channel. Our analysis suggests that by increasing W, the manufacturer induces the retailer to promote webrooming and showrooming.

However, a drop-shipping channel may not be so useful in the case of surplus production. In that case, because of relatively low W, the retailer already has an advantage and may not benefit significantly from the collaboration.

C. Sensitivity Analysis on Demand Parameters

Assuming the values of W and θ were fixed before the season starts, we carried out several analyses. Market variables like market size, attraction towards a channel, price sensitivities, and lead time sensitivities are susceptible to change. The purpose of this section is to reflect upon the inherent uncertainties in such market variables. Keeping W and θ constant at 135 and 20% respectively, we present the findings from the analyses in the following section.

Effect of increased intrinsic attractions:

Fig. 8 presents the effect of an increase in the values of intrinsic attractions α_{bmz} , α_{emz} , α_{emz} , and α_{bz} on manufacturer's profit, retailer's profit, production, and delivery lead times.



Fig. 8: Effect of increase in intrinsic attractions on profitabilities, production and lead time.

When the drop-shipping channel became more attractive, i.e., when α_{bmz} was increased by two points, the manufacturer's profit increased marginally while the retailer's profit increased by 8%. There is no change in production per unit time, but delivery lead time increased on average. It is counter-intuitive to see the manufacturer's profit increase only marginally. Also, this is the only scenario where delivery lead time increased as though the manufacturer were trying to make this channel less attractive.

When α_{emz} was increased by two points, i.e., if the manufacturer's direct channel becomes slightly more attractive, the retailer's profit decreased by 7%, production per unit time decreased by 2%, and 28

manufacturer's profit increased by approximately 2%. This is the only scenario where production quantity per unit time decreased marginally. This suggests that the manufacturer can extract more profit from fewer sales when the direct channel is more attractive.

When either α_{erz} or α_{bz} were increased by two points, i.e., when retailer's channels became relatively more attractive, the retailer's profit increased by 14% and 20%, respectively; production per unit time increased by 6% and 4%, respectively; manufacturer's profit increased marginally by 4% and 2%, respectively. In both scenarios, delivery lead time decreased on average. This suggests that when a retailer's outlets become more attractive, the manufacturer may try to improve its service in terms of how fast it can deliver the product to the customer.

From Fig. 8, it can be inferred that a) the retailer's profit decreases when the direct channel becomes more attractive, and the manufacturer's dependence on the retailer decreases and it can generate more profit for itself; and b) if there is an increase in attraction towards drop-shipping channel, the retailer benefits more than the manufacturer; the manufacturer can make this channel less attractive by increasing delivery lead time.

Effect of decrease in price sensitivities:

Fig. 9 presents the effect of the decrease in price sensitivities β_{bmz}^p , β_{emz}^p , β_{emz}^p , β_{erz}^p , and β_{bz}^p on manufacturer's profit, retailer's profit, production, and delivery lead times.



Fig. 9: Effect of decrease in price sensitivities on profitabilities, production and lead time.

When customers are assumed to be less sensitive to the drop-shipping channel prices, i.e., when β_{bmz}^{p} was reduced by 4%, the manufacturer's profit increased by 10% and the retailer's profit increased by

about 50%. The delivery lead time increased on average. The results are identical to the case when there was an increase in intrinsic attraction towards the drop-shipping channel.

When customers are less sensitive to the direct channel prices, i.e., when β_{emz}^p was reduced by 4%, the manufacturer's profit increased by 20% and the retailer's profit decreased by 45%. However, this case is slightly different from the case in which the intrinsic attraction was increased for this channel. Here, production per unit time and delivery lead time has increased on an average. We can concur that a manufacturer can relax delivery times on account of decreased price sensitivities. Nevertheless, decreased price sensitivities lead to higher demand.

When either β_{erz}^p or β_{bz}^p are decreased by 4%, we see a sharp jump in the retailer's profits. However, the manufacturer's profit decreases marginally in both cases. There is no significant change in delivery lead times, and there is only a marginal increase in production per unit time. This result is slightly different from what we observed when intrinsic attraction towards the retailer's channels was increased. The retailer's profit has increased, but there is no significant change in production. Also, the manufacturer's profit decreases instead of increasing. We can say that when β_{erz}^p or β_{bz}^p increases, a retailer has an advantage, as it can generate more profit for itself without increasing the sales volume.

From Fig. 9, it can be inferred that a) when attraction towards retailer's channels increases, the manufacturer's profit increases (see Fig. 8), whereas when price sensitivity towards retailer's channels decreases, the manufacturer's profit decreases; b) decreased price sensitivity towards drop-shipping channel benefits the retailer more than the manufacturer; and c) on account of decreased price sensitivities, higher overall demand can be expected.

Effect of increase in lead time sensitivities:

In Fig. 10, we present how the manufacturer's profit, retailer's profit, production, and delivery lead times changed when delivery lead-time sensitivities were increased. The results are obvious as when the sensitivities increased, the delivery lead time decreased significantly. Profitabilities in the supply chain decreased due to lesser demand and so did production. When lead-time sensitivity towards drop-shipping was relatively greater (i.e., β_{bmz}^{l} =0.75 and β_{emz}^{l} =0.5), the retailer's profit decreased by approximately 5%. Otherwise, the retailer's profit did not decrease significantly when the sensitivities were increased.



Fig. 10: Effect of increase in lead-time sensitivities on profitabilities, production and lead time.

VI. DISCUSSION

It is important to underline that many of the results obtained in the numerical experiments section are controlled by demand parameters that we chose at the start of the analysis. Interdependencies between profitabilities, production quantity, intrinsic attractions, price sensitivities and lead-time sensitivities can be too complex to unequivocally declare the existence of any specific pattern. That said, our analysis in Section A of the numerical experiments strongly suggests that the inclusion of a drop-shipping channel in the channel mix can assist in generating higher returns apart from serving as an omnichannel service. Additionally, there is a scope for coordination between the two players with respect to determining the value for θ , assuming the players are interested in maximizing the total supply-chain profit.

From the analysis in Section B of the numerical experiments, we can infer that adding a dropshipping channel may become an appealing option in cases when production capacity is not sufficient—that is, in the case of under-production, the manufacturer can obtain better returns using a drop-shipping channel. However, a drop-shipping channel may not be so useful in the case of surplus production. Because of over-production, the manufacturer is forced to keep *W* relatively low. The retailer already has an advantage and may not benefit significantly from the collaboration.

In the analyses we carried out in Section C, we consistently found that the increased popularity of the drop-shipping channel benefits the retailer more than the manufacturer. The results also indicate that the drop-shipping channel cannibalizes the direct channel because of the non-differentiability of prices in both these channels. We recommend that a manufacturer operating a direct channel consider these results before opting for full-scale integration of a drop-shipping channel.

Although an array of sensitivity analyses could be carried out, we rest our case highlighting that our primary aim was to draw researchers' attention towards revenue management for manufacturers operating in omnichannel retailing environments. We explored how Stackelberg equilibria can be obtained when objective functions are non-convex. The likeness of the insights generated in this paper with the recent publication [48], in which a manufacturer–retailer problem was solved using a traditional approach, strengthens the utility of the proposed bilevel programming for revenue management problems.

Limitations and Future Scope:

In the section on comparative analysis, Figs. 4a–4b shows the time complexity of the hierarchical bilevel optimization technique. As the number of variables increased, the algorithms took exponentially longer to converge. Therefore, for larger problem instances, hierarchical optimization techniques are not suitable. In the past, researchers have also used population-based solution techniques such as the genetic algorithm (GA), particle swarm intelligence (PSO), and several co-evolutionary approaches. We anticipate that those algorithms may take less time. Another approach to reducing the time complexity of multi-location larger problem instances is to convexify or linearize the retailer's objective function. This paper presented how a single-location problem can be simplified and solved using a single-level reduction technique. In the future, researchers may seek to do the same for multi-zonal problems.

Drop-shipping is a sustainable alternative when time sensitivities are not very high. We can see that when time sensitivities are high, the manufacturer seeks to expedite the deliveries, thereby leading to higher delivery costs. It is understood that when time sensitivities are too high, buying online and picking from store (BOPS) [4] and shipping from store (SFS) [49] are suitable alternatives. Extending the research further, it would be interesting to investigate how the players would collaborate when the manufacturer is ready to fulfill direct channel orders via BOPS and SFS routes, and how these facilities would affect profitabilities in the supply chain. Another avenue of research may examine how a retailer would price the

product, given that a consumer would visit the store to collect the package, or whether the manufacturer should match the price posted by the retailer.

In this paper, we assumed that the demand is deterministic and that its parameters do not change over time. This assumption is restrictive as it can be difficult to have accurate information on actual demand well before the selling season starts. This calls for dynamic bilevel optimization that proactively optimizes production, prices and fulfillments based on recent demand and market trends. To cater to uncertainties, the same problem can also be reformulated using stochastic programming instead of assuming demand to be deterministic. Finally, we assumed the manufacturer to be the leader; however, the scenario where a retailer is the leader has not been considered. The paper focused on profit maximization for both retailer and manufacturer; however, there are scenarios where retailers follow "everyday low pricing" with little regard for profit maximization in the short term.

VII. CONCLUSION

This paper proposed a joint production, pricing and inventory control problem for manufacturers operating in a competitive omnichannel environment; the premise is revenue management for manufacturers. An increasing number of manufacturers are operating direct channels to have immediate access to the market and to reduce their dependency on retailers. Direct channels play a crucial role in generating profits in the supply chain. However, while formulating a revenue management problem for a manufacturer, the direct channel must not be seen to be independent of the retailer's online and offline channels. A mere suggestion for revenue management for the manufacturers leads to bilevel optimization that essentially relates to Stackelberg leader–follower games. Keeping in view that drop-shipping may become a necessary omnichannel service, the paper simultaneously investigated the possibility of collaboration between the two players via a drop-shipping channel.

To solve the formulated problem, the paper first examined the single-level reduction technique commonly used in bilevel optimization literature. For single-level reduction, it was necessary to convexify the lower-level problem so that the necessary KKT conditions could be applied. In this paper, the lower-level problem was convexified assuming the retailer operated only one physical store. Even after reducing the bilevel problem to a single-level problem, the overall problem remained non-linear and non-convex. As

shown in the paper, the resulting problems can be solved directly using stochastic search algorithms. Nevertheless, research is required to convexify the lower-level problem if the retailer operates from multiple locations. For now, the paper proposed a hierarchical optimization technique coupled with stochastic search algorithms to solve such problems.

The paper carried out numerical experiments to generate some insights as presented in the Discussion section. Our results suggest that in the absence of a drop-shipping channel, the players would operate in an entirely competitive mode, leading to suboptimal supply-chain profit. Therefore, the inclusion of a drop-shipping channel in the channel mix not only serves as an omnichannel feature but can also assist in generating a higher profit. The paper shows how, in the presence of drop-shipping services, retailers' existing online and offline channels can effortlessly become webrooms and showrooms, respectively. The paper further suggested that adding a drop-shipping channel can a particularly appealing option in cases when the production capacity is not sufficient. However, the results also suggest that the introduction of a drop-shipping channel benefits the retailer more than the manufacturer.

APPENDIX

Proof for Proposition 1:

In the single-zone case (Z = 1), it is assumed that the retailer has an offline store and an online channel (that caters to the demand generated locally). Let $\frac{A_b^t}{1 + A_b^t + A_{er}^t + A_{bm}^t} = \lambda_b^t$, $\frac{A_{er}^t}{1 + A_b^t + A_{er}^t + A_{bm}^t} = \lambda_{er}^t$ and

 $\frac{A_{bm}^{t}}{1+A_{b}^{t}+A_{er}^{t}+A_{em}^{t}+A_{bm}^{t}} = \lambda_{bm}^{t}$ where λ_{b}^{t} , λ_{er}^{t} and λ_{bm}^{t} represent purchase probabilities at time *t* in the store, retailer's online channels and drop-shipping channels, respectively. For simplicity of expression, let us assume that holding costs are negligible and T = 1 such that the retailer's objective function and constraints (Eqs. 10–14) appear as follows:

Maximize:
$$(P_b - W)S_b + (P_{er} - W)S_{er} + (P_{em} - c_m - c_l)\theta S_{bm} - c_{er}S_{er}$$
 (A.1)

Subject to: $S_b \le M\lambda_b$ (A.2)

$$S_{er} \le M\lambda_{er}$$
 (A.3)

$$S_b, S_{er} \ge 0 \tag{A.4}$$

$$P_{er}, P_b \in \Omega \tag{A.5}$$

Next, we employ reformulation and constraint relaxation. It can be proved that P_{em} =

$$\frac{1}{\beta_{bm}^{p}} \left(\alpha_{bm}^{p} - \ln \left(\frac{\lambda_{bm}}{1 - \lambda_{b} - \lambda_{er} - \lambda_{em} - \lambda_{bm}} \right) - \beta_{bm}^{l} l_{bm} \right), \quad P_{b} = \frac{1}{\beta_{b}^{p}} \left(\alpha_{b}^{p} - \ln \left(\frac{\lambda_{b}}{1 - \lambda_{b} - \lambda_{er} - \lambda_{em} - \lambda_{bm}} \right) \right) \quad \text{and} \quad P_{er} = \frac{1}{\beta_{b}^{p}} \left(\alpha_{b}^{p} - \ln \left(\frac{\lambda_{b}}{1 - \lambda_{b} - \lambda_{er} - \lambda_{em} - \lambda_{bm}} \right) \right)$$

 $\frac{1}{\beta_{er}^{p}} \left(\alpha_{er}^{p} - \ln \left(\frac{\lambda_{er}}{1 - \lambda_{b} - \lambda_{er} - \lambda_{em} - \lambda_{bm}} \right) \right).$ Furthermore, assuming that all the demand generated at the retailer's

end is satisfied, constraints (A.2) and (A.3) can be eliminated such that $S_b = M\lambda_b$ and $S_{er} = M\lambda_{er}$. By doing so, constraint (A.4) is also eliminated because of the non-negativity of λ_b and λ_{er} . Finally, nondifferentiability due to the discrete nature of the decision variables (such as price) is also neglected. Thus, the retailer's objective function, i.e., (A.1), becomes (A.6), with no constraints to satisfy.

Maximize:
$$\Pi_{r} = \left(\frac{1}{\beta_{b}^{p}}\left(\alpha_{b}^{p} - \ln\left(\frac{\lambda_{b}}{1 - \lambda_{b}} - \lambda_{er} - \lambda_{bm}\right)\right) - W\right)M\lambda_{b} + \left(\frac{1}{\beta_{er}^{p}}\left(\alpha_{er}^{p} - \ln\left(\frac{\lambda_{er}}{1 - \lambda_{b}} - \lambda_{er} - \lambda_{bm}\right)\right) - W\right)M\lambda_{er} + \left(\frac{1}{\beta_{bm}^{p}}\left(\alpha_{bm}^{p} - \ln\left(\frac{\lambda_{bm}}{1 - \lambda_{b}} - \lambda_{er} - \lambda_{bm}\right) - \beta_{bm}^{l}l_{bm}\right) - c_{m} - c_{l}\right)\theta M\lambda_{bm} - c_{er}M\lambda_{er}$$
(A.6)

Eigenvalues for first, second and third terms can be calculated as $\left\{0, -\frac{M(\lambda_b^2 + (-1 + \lambda_{em} + \lambda_{bm} + \lambda_{er})^2)}{\beta_b^p \lambda_b (-1 + \lambda_b + \lambda_{em} + \lambda_{bm} + \lambda_{er})^2}\right\},$ $\left\{0, -\frac{M(\lambda_{er}^2 + (-1 + \lambda_b + \lambda_{em} + \lambda_{bm} + \lambda_{er})^2)}{\beta_{er}^p \lambda_{er} (-1 + \lambda_b + \lambda_{em} + \lambda_{bm} + \lambda_{er})^2}\right\},$ which are all negative semidefinite. This

means that first, second and third terms are concave in purchase probability space $(\lambda_{er}, \lambda_b)$. The fourth term is linear. Since a non-negative weighted sum of concave functions is itself concave, the retailer's objective function, when the retailer is operating in just one market zone, is proved to be concave.

ACKNOWLEDGMENT

The research was supported by the Indian Institute of Technology, Kharagpur.

REFERENCES

[1] K. T. Talluri and G. J. Van Ryzin, *The Theory and Practice of Revenue Management*, vol. 68.

Boston, MA: Springer US, 2004.

- [2] L. Dong, P. Kouvelis, and Z. Tian, "Dynamic Pricing and Inventory Control of Substitute Products," *Manuf. Serv. Oper. Manag.*, vol. 11, no. 2, pp. 317–339, Apr. 2009, doi: 10.1287/msom.1080.0221.
- [3] F. Ali, "Omnichannel fuels Target's 145% growth in ecommerce," *www.digitalcommerce360.com*, 2021. https://www.digitalcommerce360.com/2021/03/02/omnichannel-fuels-targets-145-growth-in-ecommerce/ (accessed Mar. 04, 2021).
- [4] A. Jain, J. D. Darbari, A. Kaul, U. Ramanathan, and P. C. Jha, "Facility Selection Model for BOPS Service for an Omnichannel Retail Chain," *IEEE Trans. Eng. Manag.*, pp. 1–14, 2020, doi: 10.1109/TEM.2020.3018912.
- [5] G. Gitter, M. Raymond, K. Robinson, and J. Wilkie, "The end of shopping's boundaries: Omnichannel personalization," *McKinsey & Company*, 2020. https://www.mckinsey.com/business-functions/marketing-and-sales/our-insights/the-end-of-shoppings-boundaries-omnichannel-personalization (accessed Mar. 28, 2021).
- [6] D. R. Bell, J. Choi, and L. Lodish, "What matters most in internet retailing," *MIT Sloan Manag. Rev.*, 2012.
- [7] W. Wang, G. Li, and T. C. E. Cheng, "Channel selection in a supply chain with a multi-channel retailer: The role of channel operating costs," *Int. J. Prod. Econ.*, vol. 173, pp. 54–65, Mar. 2016, doi: 10.1016/j.ijpe.2015.12.004.
- [8] A. Arya, B. Mittendorf, and D. E. M. Sappington, "The Bright Side of Supplier Encroachment," *Mark. Sci.*, vol. 26, no. 5, pp. 651–659, 2007, doi: 10.1287/mksc.1070.0280.
- [9] F. Gao and X. Su, "Omnichannel Retail Operations with Buy-Online-and-Pick-up-in-Store," *Manage. Sci.*, vol. 63, no. 8, pp. 2478–2492, Aug. 2017, doi: 10.1287/mnsc.2016.2473.
- P. Harsha, S. Subramanian, and J. Uichanco, "Dynamic Pricing of Omnichannel Inventories," *Manuf. Serv. Oper. Manag.*, vol. 21, no. 1, pp. 47–65, Jan. 2019, doi: 10.1287/msom.2018.0737.
- [11] P. Harsha, S. Subramanian, and J. Uichanco, "Omni-channel revenue management through 36

integrated pricing and fulfillment planning," 2016.

- P. Harsha, S. Subramanian, and M. Ettl, "A Practical Price Optimization Approach for Omnichannel Retailing," *INFORMS J. Optim.*, vol. 1, no. 3, pp. 241–264, Jul. 2019, doi: 10.1287/ijoo.2019.0018.
- [13] Y. S. Huang, C. H. Ho, and C. C. Fang, "Information sharing in the supply chains of products with seasonal demand," *IEEE Trans. Eng. Manag.*, vol. 64, no. 1, pp. 57–69, 2017, doi: 10.1109/TEM.2016.2623327.
- [14] A. Naimi Sadigh, M. Mozafari, and B. Karimi, "Manufacturer-retailer supply chain coordination: A bi-level programming approach," *Adv. Eng. Softw.*, vol. 45, no. 1, pp. 144–152, Mar. 2012, doi: 10.1016/j.advengsoft.2011.09.008.
- [15] R. Metters, "General rules for production planning with seasonal demand," *Int. J. Prod. Res.*, vol. 36, no. 5, pp. 1387–1399, 1998, doi: 10.1080/002075498193381.
- [16] O. Amirtaheri, M. Zandieh, B. Dorri, and A. R. Motameni, "A bi-level programming approach for production-distribution supply chain problem," *Comput. Ind. Eng.*, vol. 110, pp. 527–537, Aug. 2017, doi: 10.1016/j.cie.2017.06.030.
- [17] K. Fahimi, S. M. Seyedhosseini, and A. Makui, "Simultaneous competitive supply chain network design with continuous attractiveness variables," *Comput. Ind. Eng.*, vol. 107, pp. 235– 250, May 2017, doi: 10.1016/j.cie.2017.03.020.
- [18] L. L. Zhang, D. U. Gang, W. U. Jun, and M. A. Yujie, "Joint production planning, pricing and retailer selection with emission control based on Stackelberg game and nested genetic algorithm," *Expert Syst. Appl.*, vol. 161, p. 113733, 2020, doi: 10.1016/j.eswa.2020.113733.
- [19] L. Brotcorne, M. Labbé, P. Marcotte, and G. Savard, "A Bilevel Model for Toll Optimization on a Multicommodity Transportation Network," *Transp. Sci.*, vol. 35, no. 4, pp. 345–358, Nov. 2001, doi: 10.1287/trsc.35.4.345.10433.
- [20] G. Brown, M. Carlyle, D. Diehl, J. Kline, and K. Wood, "A Two-Sided Optimization for Theater Ballistic Missile Defense," *Oper. Res.*, vol. 53, no. 5, pp. 745–763, Oct. 2005, doi: 10.1287/opre.1050.0231.
 37

- [21] Z. Lukač, K. Šorić, and V. V. Rosenzweig, "Production planning problem with sequence dependent setups as a bilevel programming problem," *Eur. J. Oper. Res.*, vol. 187, no. 3, pp. 1504–1512, Jun. 2008, doi: 10.1016/j.ejor.2006.09.029.
- [22] H. Küçükaydin, N. Aras, and I. Kuban Altınel, "Competitive facility location problem with attractiveness adjustment of the follower: A bilevel programming model and its solution," *Eur. J. Oper. Res.*, vol. 208, no. 3, pp. 206–220, Feb. 2011, doi: 10.1016/j.ejor.2010.08.009.
- [23] A. Kheirkhah, H. Navidi, and M. Messi Bidgoli, "Dynamic Facility Layout Problem: A New Bilevel Formulation and Some Metaheuristic Solution Methods," *IEEE Trans. Eng. Manag.*, vol. 62, no. 3, pp. 396–410, 2015, doi: 10.1109/TEM.2015.2437195.
- [24] S. W. Chiou, "A bi-level decision support system for uncertain network design with equilibrium flow," *Decis. Support Syst.*, vol. 69, pp. 50–58, 2015, doi: 10.1016/j.dss.2014.12.004.
- [25] E. Ruano-Daza, C. Cobos, J. Torres-Jimenez, M. Mendoza, and A. Paz, "A multiobjective bilevel approach based on global-best harmony search for defining optimal routes and frequencies for bus rapid transit systems," *Appl. Soft Comput.*, vol. 67, pp. 567–583, Jun. 2018, doi: 10.1016/j.asoc.2018.03.026.
- [26] Y. Li *et al.*, "Optimal scheduling of isolated microgrid with an electric vehicle battery swapping station in multi-stakeholder scenarios: A bi-level programming approach via real-time pricing," *Appl. Energy*, vol. 232, no. September, pp. 54–68, 2018, doi: 10.1016/j.apenergy.2018.09.211.
- [27] H. Von Stackelberg, *The theory of the market economy*. Oxford University Press, 1952.
- Y. Chen and M. Florian, "The Nonlinear Bilevel Programming Problem: Formulations, Regularity and Optimality Conditions," *Optimization*, vol. 32, no. 3, pp. 193–209, 1995, doi: 10.1080/02331939508844048.
- [29] A. Sinha, P. Malo, and K. Deb, "A Review on Bilevel Optimization: From Classical to Evolutionary Approaches and Applications," *IEEE Trans. Evol. Comput.*, vol. 22, no. 2, pp. 276–295, Apr. 2018, doi: 10.1109/TEVC.2017.2712906.
- [30] S. T. Peinkofer, T. L. Esper, R. J. Smith, and B. D. Williams, "Assessing the impact of dropshipping fulfilment operations on the upstream supply chain," *Int. J. Prod. Res.*, vol. 57, no. 11, 38

pp. 3598-3621, Jun. 2019, doi: 10.1080/00207543.2018.1543973.

- [31] J. Huang, M. Leng, and M. Parlar, "Demand functions in decision modeling: A comprehensive survey and research directions," *Decis. Sci.*, vol. 44, no. 3, pp. 557–609, 2013, doi: 10.1111/deci.12021.
- [32] G. Vulcano, G. van Ryzin, and W. Chaar, "OM Practice Choice-Based Revenue Management: An Empirical Study of Estimation and Optimization," *Manuf. Serv. Oper. Manag.*, vol. 12, no. 3, pp. 371–392, Jul. 2010, doi: 10.1287/msom.1090.0275.
- [33] J. P. Dubé *et al.*, "Structural Applications of the Discrete Choice Model," *Mark. Lett.*, vol. 13, no. 3, pp. 207–220, 2002, doi: 10.1023/A:1020266620866.
- [34] J. D. Lemp, K. M. Kockelman, and P. Damien, "The continuous cross-nested logit model: Formulation and application for departure time choice," *Transp. Res. Part B Methodol.*, vol. 44, no. 5, pp. 646–661, 2010, doi: 10.1016/j.trb.2010.03.001.
- [35] D. Dzyabura and S. Jagabathula, "Offline Assortment Optimization in the Presence of an Online Channel," *Manage. Sci.*, vol. 64, no. 6, pp. 2767–2786, Jun. 2018, doi: 10.1287/mnsc.2016.2708.
- [36] G. Baltas and P. Doyle, "Random utility models in marketing research: A survey," J. Bus. Res., vol. 51, no. 2, pp. 115–125, 2001, doi: 10.1016/S0148-2963(99)00058-2.
- [37] A. G. Kök and M. L. Fisher, "Demand Estimation and Assortment Optimization Under Substitution: Methodology and Application," *Oper. Res.*, vol. 55, no. 6, pp. 1001–1021, 2007, doi: 10.1287/opre.1070.0409.
- [38] S. Subramanian, "Demand Modeling in the Presence of Unobserved Lost Sales," 2015.
- [39] G. Hua, S. Wang, and T. C. E. Cheng, "Price and lead time decisions in dual-channel supply chains," *Eur. J. Oper. Res.*, vol. 205, no. 1, pp. 113–126, 2010, doi: 10.1016/j.ejor.2009.12.012.
- [40] Y. Wang, H. Li, and C. Dang, "A New Evolutionary Algorithm for a Class of Nonlinear Bilevel Programming Problems and Its Global Convergence," *INFORMS J. Comput.*, vol. 23, no. 4, pp. 618–629, Nov. 2011, doi: 10.1287/ijoc.1100.0430.
- [41] A. Sinha, P. Malo, and K. Deb, "Evolutionary algorithm for bilevel optimization using approximations of the lower level optimal solution mapping," *Eur. J. Oper. Res.*, vol. 257, no. 2, 39

pp. 395-411, Mar. 2017, doi: 10.1016/j.ejor.2016.08.027.

- [42] W. Hanson and K. Martin, "Optimizing Multinomial Logit Profit Functions," *Manage. Sci.*, vol. 42, no. 7, pp. 992–1003, Jul. 1996, doi: 10.1287/mnsc.42.7.992.
- [43] S. Kirkpatrick, C. D. Gelatt, and M. P. Vecchi, "Optimization by Simulated Annealing," *Science* (80-.)., vol. 220, no. 4598, pp. 671–680, May 1983, doi: 10.1126/science.220.4598.671.
- [44] K. Mihić, K. Ryan, and A. Wood, "Randomized Decomposition Solver with the Quadratic Assignment Problem as a Case Study," *INFORMS J. Comput.*, vol. 30, no. 2, pp. 295–308, May 2018, doi: 10.1287/ijoc.2017.0781.
- [45] X. Li, P. Tian, and X. Min, "A Hierarchical Particle Swarm Optimization for Solving Bilevel Programming Problems," in *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, vol. 4029, no. 1, 2006, pp. 1169–1178.
- [46] K. H. Sahin and A. R. Ciric, "A dual temperature simulated annealing approach for solving bilevel programming problems," *Comput. Chem. Eng.*, vol. 23, no. 1, pp. 11–25, 1998, doi: 10.1016/S0098-1354(98)00267-1.
- [47] J.-Y. Lee and R. K. Cho, "Contracting for vendor-managed inventory with consignment stock and stockout-cost sharing," *Int. J. Prod. Econ.*, vol. 151, pp. 158–173, May 2014, doi: 10.1016/j.ijpe.2013.10.008.
- [48] Q. Ding, C. Dong, and Z. Pan, "A hierarchical pricing decision process on a dual-channel problem with one manufacturer and one retailer," *Int. J. Prod. Econ.*, vol. 175, pp. 197–212, 2016, doi: 10.1016/j.ijpe.2016.02.014.
- [49] D. Yang and X. Zhang, "Omnichannel operations with ship-from-store," *Oper. Res. Lett.*, doi: 10.1016/j.orl.2020.03.009.